Author's response to reviews

Title: Early menopause, association with tobacco smoking, coffee consumption and other lifestyle factors: a cross-sectional study

Authors:

Thea F Mikkelsen (theam@studmed.uio.no)
Sidsel Graff-Iversen (sgri@fhi.no)
Johanne Sundby (johanne.sundby@medisin.no)
Espen Bjertness (espen.bjertness@medisin.no)

Version: 4 Date: 1 July 2007

Author's response to reviews:

The title has been changed to "Early menopause, association with tobacco smoking, coffee consumption and other lifestyle factors: a cross-sectional study" and the keywords are removed. Best regards, Thea Falkenberg Mikkelsen